Author's response to reviews

Title: Are marginalized women being left behind? A population-based study of institutional deliveries in Karnataka, India

Authors:

Paul C Adamson (paul.c.adamson@gmail.com)
Karl Krupp (karl_krupp@phrii.org)
Bhavana Niranjankumar (bhavanaskumar@gmail.com)
Alexandra H Freeman (jana.h.freeman@gmail.com)
Mudassir Khan (mudkhan@gmail.com)
Purnima Madhivanan (pmadhivanan2@gmail.com)

Version: 4 Date: 11 January 2012

Author's response to reviews:

Dear Editor,

The galley proof had links that were not working, so we have corrected all those links and are uploading the corrected final version of the manuscript. There were no other changes made to the manuscript other than correcting the hyperlinks that had errors.

Thanks

Purnima